Author's response to reviews

Title: Factors influencing the consumption of seafood among young children in Perth: a qualitative study

Authors:

Alexandra McManus Dr (A.Mcmanus@curtin.edu.au)
Sharyn K Burns (S.Burns@curtin.edu.au)
Peter A Howat Professor (P.Howat@curtin.edu.au)
Lisa Cooper (L.Cooper@curtin.edu.au)
Lynda Fielder (L.Fielder@curtin.edu.au)

Version: 5 Date: 8 May 2007

Author's response to reviews: see over
8 May 2007

Iratxe Puebla
Senior Assistant Editor
BMC-series journals

Dear Iratxe

MS: 202038045127718 Factors influencing the consumption of seafood among young children in Perth: a qualitative study. A McManus, SK Burns, PA Howat, L Cooper, L Fielder

Thank you for the positive comments received by the reviewers of the abovementioned article. It is gratifying to note that two of the three reviewers have agreed for the article to be published.

I have revised the article to accommodate the additional comments from the third reviewers and resubmitted electronically. The specific responses to reviewer comments are listed below.

Yours sincerely

Dr Alexandra McManus

Corresponding Author
Dr Alexandra McManus
Senior Research Fellow
Assoc. Director, WA Centre for Health Promotion Research
Curtin University of Technology
PO Box U 1987
Perth, WA 6845
Phone: +61 08 9266 2115
Fax: +61 08 9266 7444
Email: A.Mcmanus@curtin.edu.au

REVIEWER: WIM VERBEKE
- Additional comments have been added to the discussion to further highlight the comparisons of this study with current literature.
- The wording of the recommendations has been altered to clarify to the readers that they are indeed recommendations for future research in this area.
- Although we note the comment from the reviewer that focus group topic guide is an alternative name for a focus group questionnaire, we have left this terminology as is given this is the recognized term in Australia where the study was conducted. Furthermore, this term is supported as the appropriate term within our University.
- Verbatim statements that are not full sentences (n=6) have now been premised with … as appropriate. Apologies for this omission in formatting.